Author's response to reviews

Title: Relationship between bone density and bone metabolism in adolescent idiopathic scoliosis

Authors:

Ko Ishida (isikoh@yokohama-cu.ac.jp)
Yoichi Aota (yaota@yokohama-cu.ac.jp)
Naoto Mitsugi (nmitsugi@yokohama-cu.ac.jp)
Motonori Kono (mo00-kohno@city.yokohama.jp)
Takayuki Higashi (t_higashi_east@yokohama-cu.ac.jp)
Takuya Kawai (seaparadise0307@yahoo.co.jp)
Katsutaka Yamada (katsutaka@af.em-net.ne.jp)
Takanori Niimura (takanoriniimura@hotmail.com)
Hironobu Tanabe (iamhiro1978@yahoo.co.jp)
Kanichiro Kaneko (kankaneko@hotmail.co.jp)
Yohei Ito (youhei1021@yahoo.co.jp)
Tomoyuki Katsuhata (winnyfielder1005@hotmail.com)
Tomoyuki Saito (t_saito@med.yokohama-cu.ac.jp)

Version: 3  Date: 28 May 2015

Author's response to reviews:

Dear Dr. Theodoros B GRIVAS and reviewers

Thank you very much for the reviewers’ reports. The criteria for menarche was the first menarche date for the information from patient and her parents and I am afraid to say that further information was not gained. The normal date for BMD from age matched control was from Kalkwarf HJ et al.[11].

The word aggravation changed to progression.